Author’s response to reviews

Title: EQUIP Emergency: Study Protocol for an Organizational Intervention to Promote Equity in Health Care

Authors:

Colleen Varcoe (colleen.varcoe@nursing.ubc.ca)
Vicky Bungay (vicky.bungay@ubc.ca)
Annette Browne (annette.browne@ubc.ca)
Erin Wilson (erin.wilson@unbc.ca)
C. Wathen (nwathen@uwo.ca)
Kat Kolar (katkolar@mail.ubc.ca)
Nancy Perrin (nperrin@jhu.edu)
Scott Comber (scott.comber@dal.ca)
Amélie Garneau (amelie.blanchet.garneau@umontreal.ca)
David Byres (David.Byres@gov.bc.ca)
Agnes Black (ABlack@providencehealth.bc.ca)
Roberta Price (priceroberta@yahoo.ca)

Version: 1 Date: 27 Aug 2019

Author’s response to reviews:

Dear Editors,

We are pleased to provide the minor revisions requested on this manuscript. Specific changes have been noted below.

1. Trial registration

We have added the name of the trial registry (Clinical Trials.gov) to this section, as requested.
2. Authors’ contributions

We have revised the contribution statement to include all authors' roles.

3. Box 1

We have converted this text box into a table format, and have included this in the revised manuscript uploaded here.

4. Figure Legend

A section entitled Figure Legends has been included after the References, and now specifies Figure 1 title. Please note that there is no figure legend to be included in this manuscript.

5. Clean manuscript

The final manuscript uploaded here has been cleaned to remove tracked changes, comments, highlights, strikethroughs and text in different colours. All tables and figures are also clean versions and have been cited within the text.

We trust that these modifications will meet with your standards. Should you require any additional changes, please do not hesitate to contact me.

Regards,

Colleen Varcoe